

Reviewer Number ID	Answering reviews
01213172	<ol style="list-style-type: none">1. Thank you for your feedback.2. Omitted the dot from the running title.3. All lymphomas are grouped into one category.4. Replaced some terms including increased risk, and populace with heightened risk and population as per reviewer feedback.5. Since the study covers a wider timeframe and incorporates both ICD-9 and ICD-10 coding systems, dividing lymphomas into distinct categories may lead to coding difficulties and errors. Consequently, our study design does not allow separating lymphomas into different categories.
03967085	<ol style="list-style-type: none">1. Thank you for your feedback.2. The risk for developing lymphoma due to celiac disease and a proposed explanation for this occurrence are included.